Author's response to reviews

Title: Fibromyalgia and neuropathic pain - differences and similarities. A comparison of 3057 patients with diabetic painful neuropathy and fibromyalgia.

Authors:

Jana Koroschetz (j.koroschetz@neurologie.uni-kiel.de)
Stefanie E Rehm (s.rehm@neurologie.uni-kiel.de)
Ulrich Gockel (ulrich.gockel@arcor.de)
Mathias Brosz (mathias.brosz@statconsult.de)
Rainer Freynhagen (r.freynhagen@krankenhaus-tutzing.de)
Thomas R Tölle (toelle@lrz.tu-muenchen.de)
Ralf Baron (r.baron@neurologie.uni-kiel.de)

Version: 3 Date: 10 May 2011

Author's response to reviews: see over
Dear Editorial Team,

Thank you very much for your positive reply. We are very grateful that the paper will be soon ready for publication. We have revised our manuscript according to the editorial comments:

- In the authors’ contributions we indicated that all authors read and approved the final version of the manuscript and we moved the ‘competing interests’ section before ‘authors’ contributions’.
- Highlighting of changes done before was removed in the entire document.
- In reference 3 and 30 the missing authors were added.
- The vertical lines in the tables were removed and the whole manuscript checked again for typographical errors.

Please do not hesitate to contact us again if you have any question concerning the revised manuscript.

Thank you very much

Sincerely yours

Jana Koroschetz, Stefanie Rehm and Ralf Baron